Author's response to reviews

Title: "If you can’t treat HPV, why test for it?" Women’s attitudes to the changing face of cervical cancer prevention - a focus group study.

Authors:

   Judith McRae (judemcrae@yahoo.com)
   Cara Martin (cara.martin@tcd.ie)
   John O’Leary (OLEARYJJ@tcd.ie)
   Linda Sharp (linda.sharp@ncri.ie)

Version: 2
Date: 26 February 2014

Author's response to reviews:

Cover letter is attached, thank you for your helpful comments.